Author’s response to reviews

Title: A case of severe hyponatremia: as a presenting feature of pituitary macroadenoma

Authors:

Kameera Bopeththa (Kameerabvkm@gmail.com)
Mohomed Niyaz (mniyas02@gmail.com)
Chathuranga Medagedara (senalmed@gmail.com)

Version: 2 Date: 09 Jan 2019

Author’s response to reviews:

Dear sir or madam

Many thanks for your valuable comments on our case report. We have done the necessary revisions. The responses for reviewers comments as follows

1. The necessary paragraph was included in the introduction page.
   Details of pre and post operative investigation results were added to the case history.
   Necessary follow up details also were included.
   CT scan photo also was added.

2. Case summary was included in the initial part of discussion.

Thank you

Kameera